Author's response to reviews

Title: Support of personalized medicine through risk-stratified treatment recommendations - an environmental scan of clinical practice guidelines

Authors:

Tsung Yu (tsyu@jhsph.edu)
Daniela Vollenweider (danivollenweider@yahoo.de)
Ravi Varadhan (rvaradhan@jhmi.edu)
Tianjing Li (tli@jhsph.edu)
Cynthia Boyd (cyboyd@jhmi.edu)
Milo A Puhan (mpuhan@jhsph.edu)

Version: 4 Date: 8 November 2012

Author's response to reviews: see over
Dear Editor
Thank you again for the favorable evaluation of our paper and the additional comments made by the reviewers. Please find below a point-by-point reply to the reviewers’ comments. We are looking forward to seeing your re-evaluation.
Sincerely,
Milo Puhan

Editor comments
1. Please rename the 'disclosure' section to 'competing interests'
   We made this change accordingly.

2. Please confirm whether figure 1 was made exclusively for this manuscript, or whether permission was obtained from the copyright holder to reproduce it in an open-access journal.
   We confirm that figure 1 was made exclusively for this manuscript.

3. Please provide a list of 3-10 keywords after the abstract.
   We actually listed keywords already but moved them to after the abstract.

Reviewer 2: only comment listed here that requires a revision
Does the manuscript adhere to the relevant standards for reporting and data deposition? as stated in my previous review, the authors could have used the PRISMA reporting guidelines which suits well an environmental scan.

We followed the PRISMA reporting guidelines where possible and uploaded the completed checklist as an online supplement.